Author’s response to reviews

Title: Hypoglycemic Encephalopathy Mimicking Acute Ischemic Stroke in Clinical Presentation and Magnetic Resonance Imaging: A Case Report

Authors:
Kai-I Chuang (kiintw@gmail.com)
Kevin, Li-Chun Hsieh (kevinh9396@gmail.com)
Cheng-Yu Chen (sandy0932@gmail.com)

Version: 2 Date: 02 Aug 2018

Author’s response to reviews:

Dear Esther Fagelson

Thank you for your comment, and we've revised the section "Consent for publication" as below:

The study protocol was approved by the Institutional Review Board of Taipei Medical University Hospital, and consent for publication was waived. (The IRB number: N201607052)

In addition, the revised manuscript was also uploaded.

Thank you very much.

Sandy, Cheng-Yu Chen, MD